Author’s response to reviews

Title: Overcoming barriers to recruiting ethnic minorities to mental health research: a typology of recruitment strategies

Authors:

Waquas Waheed (waquas.waheed@lancashirecare.nhs.uk)
Adrine Woodham (adrine.woodham@manchester.ac.uk)
Adwoa Hughes-Morley (adwoa.hughes-morley@manchester.ac.uk)
Gill Brown (gillian.brown@edgehill.ac.uk)
Peter Bower (peter.bower@manchester.ac.uk)

Version: 6 Date: 24 March 2015

Author’s response to reviews:

Dear Editor
Thank you very much for providing us the opportunity to revise our paper.

Editor’s Comments: The papers is much improved and scientifically sound. However, the quality and style of writing remains problematic. I made many edits to the paper for English, but it requires many more.

Authors reply: We have further worked on the quality and writing style of the manuscript. Changes have been where necessary particularly in introduction and discussion sections by experienced native English speaking co authors.

Best wishes
Dr Waquas Waheed